Author's response to reviews

Title: Endometriosis in postmenopausal woman without previous hormonal therapy.

Authors:

manuel garcia-manero (mgmanero@unav.es)
pedro royo (proyo@alumni.unav.es)
begoña olartecochea (bolarteco@unav.es)
juan luis alcazar (jlalcazar@unav.es)

Version: 2 Date: 7 January 2009

Author's response to reviews: see over
To The Editor of “Journal of Medical Case Reports”:

December 30th, 2008

Dear Editor,

Enclosed you will find a new manuscript to consider its revision and possible publication in your journal. The manuscript entitled “Endometriosis in postmenopausal woman without previous hormonal therapy: a case report” is an Original Article.

We certify that this paper has not been submitted simultaneously not is under revision in any other journal and that high standards of ethics have been applied carrying on the investigations.

Sincerely yours,
Manuel García-Manero, PhD

Corresponding author:
Manuel García-Manero, PhD
Department of Obstetrics and Gynecology
Clinica Universitaria de Navarra. University of Navarra.
Avenida Pío XII, 36
31008 Pamplona
Spain
Fax: 34-948-296500
e-mail: mgmanero@unav.es